Author’s response to reviews

Title: Menstrual Health and Hygiene among Indigenous Australian girls and women: Barriers and opportunities

Authors:

Emily Krusz (e.krusz@uq.edu.au)
Nina Hall (n.hall2@uq.edu.au)
Dani Barrington (d.j.barrington@leeds.ac.uk)
Sandra Creamer (s.creamer@uq.edu.au)
Wendy Anders (seniorprojects@natsiwa.org.au)
Minnie King (womenoncountry@bigpond.com)
Helen Martin (womenoncountry1@bigpond.com)
Julie Hennegan (jhenneg1@jhu.edu)

Version: 2 Date: 28 Aug 2019

Author’s response to reviews:

Dear Editor,

Thank you for your helpful feedback.

I can confirm that this article now confirms to a ‘Debate’-style article for BMC Women’s Health after revisions and checking.

This manuscript does not contain any primary research data. It draws on two previous publications by the co-authors, and we have amended the article to be clearer about this, conforming to the requirements of a Debate article.

We have provided two versions: a ‘tracked changes’ copy and a ‘clean’ copy to clearly identify the places in the article that have been revised.
Please let us know if further information is required.

Sincerely,

Nina Hall, on behalf of the author team